

MDPI

Review

Financial Inclusion and Its Ripple Effects on Socio-Economic Development: A Comprehensive Review

Deepak Mishra 1, Vinay Kandpal 2,3,* D, Naveen Agarwal 4 and Barun Srivastava 1,5

- School of Business, UPES, Dehradun 248007, India; deepak.107510@stu.upes.ac.in (D.M.); barun.2189@gmail.com (B.S.)
- ² Department of Management Studies, Graphic Era Deemed to Be University, Dehradun 248002, India
- ³ Graduate Program in Business Administration, Paulista University, UNIP, São Paulo 04026-002, Brazil
- ⁴ Centre for Continuing Education (UPESON), UPES, Dehradun 248007, India; naveen.agarwal@ddn.upes.ac.in
- ⁵ School of Commerce, Graphic Era Hill University, Dehradun 248002, India
- Correspondence: vinayaimca@gmail.com

Abstract: This study provides an overview of the different dimensions of financial inclusion, its socioeconomic impacts on society's sustainable development, and future research agendas. Initially, 620 studies were identified using Scopus and other databases, employing keywords such as financial literacy, financial inclusion, financial capability, women's empowerment, fintech, artificial intelligence, financial accessibility, sustainable development goals, and economic growth. After refinement based on focus and relevance, 325 papers were analyzed in detail for review, primarily focused on India and emerging economies. This review highlights that access to finance by untouched segments of society is essential for sustainable and socio-economic development in developing economies. The official banking system, an effort by the government to assist the financially disadvantaged, can incorporate the impoverished into a formal financial system through campaigns and credit system reforms. Socioeconomic programs reinforce one another and foster the development of children, women, families, and society. This research paper undertakes a systematic literature review primarily focused on relevant articles in broad areas of financial inclusion and its impact analysis and offers a valuable agenda for future research.

Keywords: financial inclusion; systematic review; banking; financial services; sustainable development; women empowerment

1. Introduction

The ability of people and businesses to obtain practical and reasonably priced financial products and services that satisfy their needs—transactions, payments, savings, credit, and insurance—delivered in an ethical and sustainable manner is known as financial inclusion. It is seen as a major force behind financial well-being and economic growth (Sapre 2023). According to recent research, financial inclusion is a multifaceted concept that includes having and using financial goods, such as mobile money accounts, as well as risk management (insurance), savings accounts, credit availability, and receiving remittances (Koomson and Ibrahim 2018; Koomson et al. 2020). According to the World Bank's Global Financial Database 2017, a sizable portion of the population remains financially excluded (Demirguc-Kunt et al. 2018). The emergence of financial inclusion as a prime development challenge has attracted researchers and policymakers worldwide (Alliance for Financial Inclusion 2019; Han and Melecky 2013). However, 31 percent of the world's population does not have access to an organized financial system (Demirguc-Kunt et al. 2018). BRICS alone accounts for half of that, India and China account for 32 percent, Brazil accounts for 2.4%, South Africa accounts for 0.5%, and Russia accounts for less than 0.5%. For women (+15 age) in India, the participation further drops to 43 percent. Just 6 percent of the total population in India borrowed from a formal financial institution in 2014 compared to 10 percent in



Citation: Mishra, Deepak, Vinay Kandpal, Naveen Agarwal, and Barun Srivastava. 2024. Financial Inclusion and Its Ripple Effects on Socio-Economic Development: A Comprehensive Review. Journal of Risk and Financial Management 17: 105. https://doi.org/10.3390/jrfm17030105

Academic Editors: Peterson K. Ozili and Thanasis Stengos

Received: 24 December 2023 Revised: 11 February 2024 Accepted: 19 February 2024 Published: 3 March 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

other BRICS economies. Furthermore, India had just 18 ATMs per 100,000 adults in 2014, compared to over 65 in South Africa and 180 in Russia. Only approximately 10 percent of adults (+15 age) used ATM cards for payments compared to 40 percent in South Africa. Moreover, formal saving and borrowing remained extremely low. Only 53 percent of individuals (+15 years) hold accounts with formal financial institutions (Reserve Bank of India 2015). Financial Inclusion helps in improving accessibility to financial products and services, such as payments, insurance, loans, and others (Babajide et al. 2015; Asongu et al. 2020). The provision of practical and reasonably priced financial services and products (such as credit, insurance, savings, and payments) that meet the requirements of individuals and companies in an ethical and long-term way is known as financial inclusion (World Bank 2018).

Providing banking and financial services at a reasonable price is important in areas where the interaction between formal economic systems and individuals (such as homes or businesses) is limited. It is risky to promote digital finance as a pro-development financing tool for reducing poverty for two reasons: first, the wealthy can more easily transfer or mitigate the risks associated with using digital finance platforms by purchasing insurance, while the impoverished lack the resources to do so (Koh et al. 2018; Ozili 2020). In addition, financial services availability provides a possible balanced consumption, plans recurring expenses effectively, and meets some emergency requirements (Babajide et al. 2015; Lauer and Lyman 2015). Financial inclusion encompasses a number of elements, such as availability, expansion usage, and impediments to financial services. A significant body of work is available to present the determinants of financial inclusion in India. For instance, branch networks (Kumar 2013), self-help groups and education (Bhanot et al. 2012); socioeconomic factors (Clámara et al. 2014; Rastogi and Ragabiruntha 2018); domestic agriculture to gross domestic product (GDP), literacy ratio, population density, and urbanization (Yadav and Sharma 2016); economic growth (Sharma 2016); bank branch quantity, and credit deposit ratio on GDP (Iqbal and Sami 2017); infrastructure (Sharma et al. 2018); poverty and inequality (Erlando et al. 2020); and social safety (Didenko et al. 2020). People are being helped to escape poverty and are being connected to economic opportunities through formal financial services, including those that may be accessed over the phone. It might be challenging for the impoverished to save money for future companies or educational investments without a savings account. A crop disaster without insurance could leave farmers and their families penniless (Klapper et al. 2016).

The literature shows that banks are the most significant drivers of the financial inclusion process in India. Information technology and communication led to financial services, and Jan Dhan-Aadhaar's mobile and essential infrastructure presence substantially affected the general conduct of financial inclusion. Many such services that are performed in India were either region-specific or an element of a broader set of proxies was lacking (Malik et al. 2020). Financial inclusion is considered a critical element in poverty reduction and economic growth policies. Some reported instances of financial inclusion success in developing nations are associated with the use of branchless banking enabled by information and communication technologies (Diniz et al. 2012). Financial technology (fintech) has upended the conventional financial system by decoupling, decentralizing, and demystifying finance. However, effective and scalable fintech adoption is essential for financial inclusion and value creation through new business models. While theoretical frameworks and models in the literature investigate technology adoption generally, there is no measurement methodology for fintech-specific customer readiness (Mahmud et al. 2023). In the insurance industry, artificial intelligence and machine learning are also widely used, particularly in risk assessment and claims reserving (Chukhrova and Johannssen 2021; Pnevmatikakis et al. 2021). The digital age and the introduction of financial technologies have enabled access to different financial products and services with a single click. However, little research to date has examined the relationship between the use of these new technologies and financial literacy (Andreou and Anyfantaki 2021). The concept of financial inclusion, which entails the accessibility and effective use of financial services, has been increasingly

recognized as a pivotal factor in socio-economic development. Despite substantial progress in this area, there remains a discernible gap in our understanding of the broader, systemic impacts of financial inclusion, particularly its 'ripple effects' on various socio-economic dimensions. Existing research predominantly focuses on the direct benefits of financial inclusion, such as enhanced financial security and poverty reduction (Demirguc-Kunt et al. 2018). However, the indirect effects, which might be equally significant, are not as thoroughly examined. This paper seeks to address this gap by posing the following research question: Beyond its immediate economic benefits, how does financial inclusion contribute to broader socio-economic development? This inquiry is vital, as it extends the understanding of financial inclusion beyond individual financial well-being to encompass wider societal and economic implications. It aligns with the call by (Park and Mercado 2018) for a deeper exploration of the long-term developmental impacts of increased access to financial services.

The relevance of this research question is further emphasized by studies like (Allen et al. 2016), who explore the relationship between financial inclusion and economic growth, hinting at a broader socio-economic impact yet leaving room for more detailed analysis. Moreover, Demirguc-Kunt et al. (2018) offer extensive data on financial inclusion but do not fully delve into its indirect socio-economic effects, presenting an opportunity for further investigation.

This paper aims to bridge this gap by providing a comprehensive review of the indirect effects of financial inclusion on socio-economic development. By doing so, it not only contributes to the academic discourse by synthesizing existing research but also identifies potential areas for future research, thereby enhancing the understanding of the multi-faceted role of financial inclusion in global development.

Financial inclusion and socioeconomic development are intricately linked, with numerous studies and literature highlighting how improved access to financial services can significantly contribute to various aspects of socioeconomic development. Below is a detailed explanation supported by the relevant literature:

Poverty Alleviation: Financial inclusion is crucial in reducing poverty, a key aspect of socioeconomic development. By providing the unbanked and underbanked populations with access to basic financial services such as savings accounts, credit, insurance, and payment services, financial inclusion enables individuals to save, invest, and protect themselves against financial risks. This empowerment leads to an increase in household income and consumption, contributing to poverty reduction. The World Bank's Global Findex Database provides extensive data on how financial inclusion is a key enabler for reducing poverty and boosting shared prosperity. The theoretical and conceptual underpinnings connecting financial inclusion to socioeconomic development attributes revolve around several key ideas:

Access to Financial Services: Financial inclusion implies that individuals and businesses have access to useful and affordable financial products and services that meet their needs—transactions, payments, savings, credit, and insurance—delivered in a responsible and sustainable way. This access is crucial for spurring economic growth, reducing poverty, and boosting prosperity.

Economic Growth: Financial inclusion is also linked to economic growth. By bringing more people into the formal financial system, it increases the pool of available capital for investment, which is crucial for economic growth. Studies have shown that when individuals and businesses have access to financial services, they can invest more in their operations, increase productivity, and contribute to overall economic growth. A report by the International Monetary Fund (IMF) highlights the positive impact of financial inclusion on economic growth.

Gender Equality: Financial inclusion plays a significant role in promoting gender equality, another important attribute of socioeconomic development. Women, who have traditionally been excluded from formal financial systems, gain more control over their finances and economic decisions when they have access to financial services. This em-

powerment leads to improvements in family well-being and education and promotes gender equality. The World Bank and various gender-focused studies emphasize the role of financial inclusion in empowering women.

Financial Literacy and Education: Improved financial inclusion often goes hand-inhand with increased financial literacy. People who are newly included in the financial system gain knowledge about managing money, saving, investing, and understanding credit. This education is a critical component of socioeconomic development as it enables individuals to make informed financial decisions. The OECD has published several reports on the importance of financial literacy for socioeconomic development.

Entrepreneurship and Job Creation: Financial inclusion supports entrepreneurship by providing individuals with access to credit and other financial services necessary to start and grow businesses. This leads to job creation, another key element of socioeconomic development. The International Labour Organization (ILO) and other studies have documented the positive correlation between financial inclusion, entrepreneurship, and job creation.

Social Stability and Reduced Inequality: By providing financial services to the lower-income and marginalized segments of society, financial inclusion helps reduce income inequality and promote social stability. Access to affordable financial services enables these groups to improve their living standards and contribute more significantly to the economy. The United Nations Development Programme (UNDP) has recognized the role of financial inclusion in achieving the Sustainable Development Goals, particularly in reducing inequality.

Innovation in Financial Services: The drive for financial inclusion spurs innovation in financial services, particularly in digital financial services like mobile banking. This innovation not only makes financial services more accessible but also more affordable, further enhancing socioeconomic development. Studies on the impact of mobile money services in various countries provide evidence of this relationship.

Women's Empowerment: Financial inclusion often leads to women's economic empowerment by giving them control over financial resources, which enhances their economic participation and decision-making power in households and communities.

Financial inclusion is a powerful catalyst for socioeconomic development, influencing various aspects such as poverty reduction, economic growth, gender equality, financial literacy, entrepreneurship, social stability, and innovation in financial services. The literature from World Bank, IMF, OECD, ILO, UNDP, and other reputable organizations provides ample evidence and insights into these connections.

Policymakers have mixed opinions about the program's success and financial inclusion. "Existing research about a marketing mix of services in the financial services industry is contradictory and confusing" (Paul et al. 2016). The current marketing framework of the banking structure is not helpful and should be restructured to use a range of accessible products. These outcomes also warrant a thorough review of the social and economic challenges of financial inclusion. Therefore, we undertake a novel systematic literature review (SLR) by analyzing publications in reputed journals to develop knowledge of financial inclusion and its socioeconomic advancement and provide further research agenda. This paper aims to shed light on the multi-dimensional impacts of financial inclusion on socio-economic development, examining both the benefits and the challenges, as well as the role of various stakeholders in facilitating this process. Based on the preceding discussions, we have formulated three focused research questions (RQs) to further explore the relationship between financial inclusion and socioeconomic development:

RQ1: Assessing Current Understanding and Knowledge: What is the current level of understanding and knowledge regarding the impact of financial inclusion on socioeconomic development? This question aims to evaluate the existing literature and studies to determine what is already known about how financial inclusion contributes to socioeconomic progress.

RQ2: Identifying Focus Areas for Socioeconomic Development: Which specific areas within socioeconomic development require more attention in the context of financial inclusion? This question seeks to identify the critical areas within socioeconomic development that are most influenced by financial inclusion and where focused efforts could yield significant benefits.

RQ3: Uncovering Research Gaps and Future Directions: What are the potential gaps in research and future directions for studying the relationship between financial inclusion and socioeconomic development? This question aims to pinpoint areas where further research is needed, identifying gaps in current knowledge, and suggesting future research directions to advance the understanding of how financial inclusion can drive socioeconomic development.

A systematic review of the literature on the various aspects of financial inclusion would help us expand existing knowledge on the assessment of financial inclusion and sustainable development. Financial inclusion is directly linked to several of the United Nations' Sustainable Development Goals (SDGs), especially those related to poverty eradication, gender equality, decent work, and economic growth.

An SLR provides comprehensive insights on how increasing access to financial services contributes to the broader goals of global development and poverty reduction. The structure of this paper includes Section 1 discussing the concept and the motivation behind the study. Section 2 discusses the methodology employed to conduct this research. Section 3 provides a theoretical context for financial inclusion, while Section 4 presents result from the literature on different themes or elements of financial inclusion. Section 5 presents the review's conclusion, and Section six presents the theoretical extension of financial inclusion and the scope of future research for inclusive growth globally.

2. Research Methods

A systematic literature review is carried out for this research paper, whose themes are sustainable development and financial inclusion. In systematic review researcher analyses, the research area covers a broad spectrum of available literature and lays down research questions that contribute to the knowledge (Tranfield et al. 2003).

Qualitative systematic reviews need to identify relevant qualitative evidence systematically. After the review collection, the studies are organized and analyzed. This research paper undertakes a systematic literature review that seeks and focuses primarily on relevant articles in broad areas of financial inclusion.

2.1. Journal Selection Criteria

The SLR technique uses the following five steps (Denyer and Tranfield 2009); the literature was scrutinized as per the issue identified above, utilizing a database. The researcher advised a set of guidelines, which was adopted for the SLR methodology in this paper. The literature was analyzed and synthesized to identify existing results, research orientations, and gaps. The articles were researched and separated from high-quality, high-impact journals indexed in electronic databases. It includes databases such as Scopus, Web of Science (WoS)/Social Science Citation Index (SSCI)/Journal Citation Report, which list academic journals with an impact factor (IF) for identifying potential sources for review (Paul et al. 2017). These are the world's premier databases for published articles and citations. When several hundred papers on a trending topic have already been published to be potentially reviewed, one can even rely upon JCR-indexed journals with an IF above a given threshold, i.e., 1.0 plus following, for instance, (Paul and Rosado-Serrano 2019). It includes publications in top-tier journals, and journals covering financial inclusion, financial literacy, women's empowerment, financial capability, fintech, and access to finance were considered for the systematic review. In order to allow researchers to begin to work in the field with a general understanding, an early SLR could provide a better-defined definition and understanding of the research field. In immature areas, review articles focus less on

hypotheses or research questions and devote more attention to the synthesis of basic facts in a given area which provides valuable insight (Kraus et al. 2020).

2.2. Article Selection Criteria

This research started by researching and reviewing the available literature on a broad area of financial inclusion from 2012 to 2023. The keywords used in the study are financial inclusion, sustainable development, self-help groups, financial technology, financial literacy, financial exclusion, artificial intelligence, and financial capability. We refined the available papers by setting the review's scope to include empirical studies, and the theoretical perspective of financial inclusion and its other related areas were considered for review work. A systematic review article can be developed using 40-50 to 500 or more relevant papers. It was decided that terms should appear in the titles of articles, abstracts, or keywords. The search was performed on 2 August 2023. During the initial phase, 620 research articles were identified. After further refinement based on focus and relevance, 325 papers were analyzed in detail for review and classified according to the type of study primarily focused on India and emerging economies. This approach aligns with our goal of integrating a substantial amount of knowledge, spotting any gaps, and suggesting future directions for study (Kitchenham et al. 2009). SLR is a rigorous methodology that generates knowledge and pinpoints a field's future research subjects, routes, and trends (Tranfield et al. 2003). The review criteria are crucial in the systematic phase. The selected papers were and understood considering the relevance of the topic.

3. Theoretical Perspective

3.1. Socioeconomic Development Theory

Tinker (2001), discussed the increased measures undertaken for socioeconomic development to uplift weaker or underprivileged women in developing countries. Activists, practitioners, and academics engage in international development policies and programs.

3.2. Theory of Planned Behavior: Understanding Behavioral Predictions

The theory of planned behavior is a psychological framework that provides insight into how behavior can be predicted based on certain key factors. This theory posits that attitudes towards conduct, subjective norms, and perceived behavioral control are integral in predicting various forms of behavior. These elements, in conjunction with one's intentions and perceptions of control over the behavior, can account for a substantial portion of the variance in actual behavior (Ajzen 1991).

The theory suggests that these three factors lead to the formation of behavioral intentions. In other words, how much a person is willing to try, and how hard they plan to work towards performing the behavior. These intentions, coupled with perceptions of behavioral control, are key predictors of actual behavior.

The theory of planned behavior (TPB) is a valuable concept in promoting financial inclusion as it helps in understanding and influencing the behavioral intentions towards using financial services. By analyzing attitudes towards financial activities, the influence of social norms, and perceived control over financial decisions, the TPB can identify barriers to financial inclusion. It aids in designing interventions to change negative perceptions, leverage social influences, and enhance the perceived ease and accessibility of financial services. This approach is crucial for developing targeted financial products, policies, and educational programs that increase participation in financial systems, especially among underserved populations. Thus, the TPB offers a comprehensive framework to address psychological and social factors critical for effective financial inclusion strategies.

The TPB is used in retail (mortgage) lending, investor education, and financial counseling, among other areas of financial sector utility (Xiao 2008). In addition to the aforementioned, British consumers were also examined using the TPB to look into investment-based decisions (East 1993). Bansal and Taylor (2002) applied it to investigate the switching behavior on the sample of mortgage loan customers. Key factors that shape consumer

behavior and intentions towards using mobile coupons are their attitudes and perceived behavioral control. In contrast, subjective norms do not show a clear influence. Within the 'behavioral attitude' category, perceived usefulness significantly impacts the attitude. Regarding 'subjective norms', the impact of the primary group is noticeable. Most importantly, under 'perceived behavioral control', self-efficacy emerges as the most influential factor (Hsu et al. 2006). The TPB helps in understanding the psychological and social factors that drive the use of financial services. By addressing these factors, interventions can be designed to enhance financial inclusion, such as through targeted financial education programs, community engagement strategies, and user-friendly financial service offerings. Understanding these behavioral predictors is crucial for developing policies and services that encourage wider participation in financial systems, particularly among underserved or excluded populations, thereby contributing to overall socio-economic development.

3.3. Technology Acceptance Model: Understanding Consumer Adoption of Technology

The technology acceptance model (TAM) is an essential theoretical framework that elucidates the process through which consumers adopt and use new technologies. This model, widely used in numerous research studies, explains that the decision-making process concerning the adoption and utilization of technology is influenced by various factors. The TAM's core value lies in its ability to help researchers and practitioners understand the reasons behind the acceptance or rejection of a technology. The TAM focuses on perceived usefulness and ease of use, which are critical in determining whether individuals will accept and continuously use technological solutions like mobile banking, online financial management tools, or digital payment systems. By understanding and enhancing these perceptions, financial institutions and policymakers can design and implement more user-friendly and beneficial financial technologies. This not only promotes wider financial inclusion, particularly among technologically underserved populations, but also contributes to socio-economic development by facilitating more efficient, accessible, and inclusive financial services, thereby improving economic activities, reducing transaction costs, and fostering financial empowerment.

The continuous evolution and integration of technology into both private and professional spheres raises an ongoing question: should a particular technology be accepted or rejected? Since its inception over a quarter-century ago, the Technology Acceptance Model has remained a dominant framework in the field of technology acceptance. It has become a key tool in understanding the predictors of human behavior in relation to the potential adoption or rejection of technological innovations. The TAM provides a comprehensive approach for analyzing the complex interplay of factors that influence technology adoption, making it an indispensable model for understanding and navigating the ever-changing landscape of technology in society (Marangunić and Granić 2015). While the TAM primarily focuses on the adoption of technology, its principles are highly applicable in understanding and enhancing the adoption of financial technologies, which is a critical aspect of financial inclusion and can have significant impacts on socio-economic development. Integrating the TAM in this context can provide a comprehensive understanding of how to effectively leverage technology to promote financial inclusion and, in turn, socio-economic growth.

4. Discussion

Financial inclusion is the most sought-after term. Existing gaps should be narrowed to ensure complete financial inclusion. All over the world, people are still struggling to close the gap and reach the unbanked. Access to financial programs and a better life are imperative for the poor and disadvantaged. This is difficult and requires all sectors' proper planning and collaboration. Financial literacy, behavior and knowledge should be more valuable in making informed decisions.

4.1. Financial Inclusion, Fintech, and Artificial Intelligence

Applications based on AI and ML have seen a notable rise in the BFSI sector, attributed to their ability to streamline processes (Moutinho and Smith 2000), refine decision-making (Polychroniou and Giannikos 2009), augment customer experiences (Zeinalizadeh et al. 2015), and identify fraudulent activities (Vorobyev and Krivitskaya 2022). Hanafizadeh and Zare Rayasan (2018) found that by outsourcing e-banking services, it is possible to categorize all 23 factors into three distinct segments: technological, environmental, and organizational. According to the author, fintech will become the standard in Asia once it is widely adopted by the majority of the people and when large institutions utilize the technology to provide their services. Enhancing financial inclusion will facilitate the integration of small enterprises, farmers, and general economic growth into the country's income. The mobilized notion of consumer interest, anchored in the rhetoric of consumer technological empowerment, transcends and conceals necessary considerations regarding the vulnerability of consumers in the context of data-intensive technologies and the platform economy (Ferrari 2022). AI and ML are widely utilized in the insurance industry, especially for tasks like claims reserving and assessing risks (Chukhrova and Johannssen 2021).

Financial inclusion in the digital mode is a primary goal for the UN Sustainable Development Goals. To achieve this, appropriate communication infrastructure is necessary and is possible in countries with a good range of communication infrastructure (Arner et al. 2020). "FinTech's and digitization have made consumers vulnerable to poorly informed financial decisions and financial fraud." Thus, there is a need for digital financial literacy for disadvantaged and vulnerable groups (Grohs-Müller and Greimel-Fuhrmann 2018). Establishing the new infrastructure for building digital infrastructure leads to data availability and links judicial processes with financial transactions (Arner et al. 2019). With the advent of information and communication technology and its association with financial platforms, a system has emerged that provides new opportunities to close the wealth disparities in developing nations. In the context of the developing region, empirical investigations are uncommon, and this study aims to cover this void (Hussain et al. 2023).

Digital transactions are comparatively low among the rural population, older people, women, and less educated people. Providing digital infrastructure to inaccessible areas will bring more awareness and financial literacy for the marginalized and the deprived, which is the first step for financial inclusion (Barik and Sharma 2019). The BFSI (Banking, Financial Services, and Insurance) sector is also embracing the advancements of AI (Artificial Intelligence) and ML (Machine Learning). These technologies are increasingly being integrated into the sector's operations, with the aim of enhancing operational efficiency, elevating the customer experience, and reducing risks (Mi Alnaser et al. 2023; Goodell et al. 2021). George and Mpeera (2020) analyzed the impact of the acceptance and usage of mobile banking on the promotion of financial inclusion. The implementation and utilization of this technology should be supported by a robust regulatory framework to effectively manage and prevent cyber fraud. (Agarwal and Chua 2020) stated that utilizing financial technology is advantageous for households, as it facilitates the augmentation of consumption and borrowing. Kawimbe (2020) found that the concept of adoption and usage of digital modes such as mobile banking, internet banking and financial literacy is still emerging. Disruptive technologies, specifically Artificial Intelligence, have permeated the retail sector and had a significant impact on organizational resources, both internal and external. Fintech startups have also been significantly impacted by the surge of digital transformation (Almansour 2023). AI and ML methods have a significant influence on improving the efficacy and efficiency of anti-money laundering initiatives in the banking, financial services, and insurance (BFSI) industry. Strong AML processes are becoming more and more important as the financial sector gets more digitalized and as internet transactions increase (Pattnaik et al. 2024).

In fintech research, the technology acceptance model (TAM) was frequently expanded to examine customer concerns. Wang (2022) conducted a comparison of verification tech-

niques in fintech, utilizing the technology acceptance model (TAM) as a basis for the analysis. The implementation of digital banking in The Kingdom of Saudi Arabia was examined by (Alnemer 2022). Bajunaied et al. (2023) investigated consumers' behavioral intention to use fintech services. Taking the aforementioned into consideration, this study uses the TAM to develop a research methodology for rearranging and assessing consumer adoption of Open API in fintech apps.

Technology businesses are responding to consumer expectations by offering more convenient and affordable ways to borrow money, invest, and transfer funds. Modern technologies, evolving consumer behavior, and the rise of fintech are all causing disruptions in the insurance sector. Fintech technologies have the potential to improve risk assessment, increase financial inclusion, save costs, and boost efficiency in the insurance sector, among other benefits (Hassan et al. 2023).

4.2. Financial and Digital Literacy of Consumers and Financial Inclusion

The authors note the paucity of research on financial literacy and financial conduct (Jappelli and Padula 2013; Lusardi et al. 2014; Lusardi and Tufano 2015; Söderberg 2016). This premise is based upon psychological reactions such as attitude, bias, and confidence, which lead to a particular behavior towards financial matters. Therefore, there needs to be an education of financial aspects to create awareness to witness positive behavior to alter the behavior. Financial literacy could be defined as "the ability of an individual to plan their finances, plan for debt and retirement and maintain wealth" (Lusardi and Mitchell 2014). High financial literacy is advantageous because it increases savings and helps in the efficient usage of funds (Wann 2017). Even after controlling income and education, individuals with higher financial literacy scores are more likely to hold formal and informal savings than those with lower financial literacy scores (Morgan and Long 2020). The depth of digital finance usage is measured by specific services used by users, like payment, credit, fund management, insurance, and investment (Xi and Wang 2023).

Kass-Hanna et al. (2022) demonstrates that both financial and digital literacy are crucial elements in fostering inclusion and financial resilience. Regions, along with destitute, rural, and families led by females, display variability. In order to address the potential issue of endogeneity, a robustness test is incorporated. The findings underscore the necessity of redefining conventional financial literacy to incorporate digital literacy, which has substantial ramifications for governments considering both as a combined approach to bolstering households' enduring financial resilience.

Less financially literate people are less likely to invest in the stock market (Mouna and Jarboui 2016). Age, education level, and annual income affect financial literacy levels. Cude et al. (2019) stated that the higher the income is, the more financial knowledge there is. The author recommended formulating a national education strategy focusing on financial literacy. "Financial literacy is an issue with vast implications for economic health, and its development can pave the way for robust and stable economies". Financial literacy increases and sharpens the economic decisions of an individual (Goyal and Kumar 2020).

Grohmann et al. (2018) stated that financial literacy leads to financial inclusion based on the S&P global financial literature survey data. Baidoo et al. (2018) noted that financial literacy level affects saving habits and identified income distribution, size of the home, and the factors that lead to savings. Adetunji and David-West (2019) surveyed more than 22,000 respondents in Nigeria and found that financial literacy affects savings patterns, whether taken from formal or informal financial institutions. Financial literacy positively impacts the inclusion and savings habits in countries such as Cambodia and Vietnam (Morgan and Trinh 2019). Policymakers need to take different initiatives and ensure that their proper implementation for society's sustainable development will improve their economic well-being (Goyal and Kumar 2020).

4.3. SHGs, Women Empowerment and Financial Inclusion

It is worth mentioning that some of the programs have been hit, while some failed due to corruption, ineffective planning, and bureaucratic mistakes. Credit facilities must be offered to improve entrepreneurship among women. The Self Help Group—Bank Linkage Programme (SHG-BLP) significantly reduces social exclusion among participants compared to non-participants and also ensures financial inclusion (Maity 2023).

Access to savings opportunities has modest but more consistent benefits for the impoverished than credit and carries fewer downside risks for clients (Duvendack and Mader 2020). The research highlights the key success factors (KSFs) that led to the GNFC Neem Project's socioeconomic success. Key success factors include maintaining a strategic alignment between social impact and economic feasibility; building market awareness, trust, and acceptance; developing win-win partnerships with diverse stakeholders; and leveraging distribution channels and networks for last-mile collection, storage, and transport of neem seeds (Goyal et al. 2020).

4.4. Intensifying Access to Financial Services to Consumers

Financial services have reached different sections of society and are tried using various methods. A sound functioning system allows formal financial institutions to channel the money into an appropriate opportunity for growth, wealth equity, and poverty reduction. Factors such as transaction and contract enforcement costs and information irregularities in the financial market significantly bind the poor to enter the financial market. Government programs that support self-help groups (SHGs) generally target women on the premise that this improves women's decision-making (Kochar et al. 2022).

As (Paramasivan and Ganeshkumar 2013) have mentioned, the greater the bank branches' density, the more we can find people using banking services. They stated that financial literacy is not the only criterion for financial inclusion but that other means, such as investment, play a crucial role in improving investment. Allen et al. (2016) reported that the formal usage of accounts determines financial inclusion and indicates gender, age, and poverty as part of the financial system. The low-cost accounts must have higher accessibility so that everybody—the poor, elderly, rural, young—can come under its ambit. Lending to women benefits their households by diversifying their incomes and enhancing their resilience to disruptions. Therefore, microcredit can be a potent tool for boosting incomes and shielding families from the risk of crises (Garikipati 2008).

4.5. Financial Inclusion and Economic Growth

Financial inclusion, which is usually understood as access to formal financial services such as credit, insurance and secure saving opportunities, has been identified as a critical engine of economic growth (Claessens and Perotti 2007). An empirical analysis was conducted to examine the impact of the coordination between agricultural insurance and digital financial inclusion on agricultural output. The study constructed a regression model to reveal that the coupling coordination degree significantly enhances farmers' agricultural output. Furthermore, the promotion effect of this coordination is more noticeable in mountainous areas and eastern China (An et al. 2023). Financial inclusion is a crucial determinant of economic growth. In South Africa, the availability of financial services led to a good life with increased savings and better financial decisions in the household (Karlan and Zinman 2010). Han and Melecky (2013) mentioned that the greater the financial reach is, the greater the stability. They added, "greater access to bank deposits can make the funding base of banks more resilient in times of financial stress". (Levine 2005; Pasali 2013) suggested that financial well-being is associated with growth and employment and has an enormous impact.

Agricultural insurance is an effective risk management tool that compensates farmers for post-disaster losses, contributing to the stability and continuity of agricultural production (Tappi et al. 2022). Ardic et al. (2013) reported the global stand on financial inclusion and found that high-income countries had ten times the deposit penetration as low-income

countries. The lower-middle-income countries had three times the deposit penetration of low-income countries. There was steady growth in the number of commercial bank branches and ATMs. Banerjee and Francis (2014) determined the relationship between social development and financial inclusion based on poverty eradication, employment generation and social harmony. Their results showed that Kerala and Maharashtra had higher financial inclusion rates, whereas Punjab and Tamilnadu scored average, but Mizoram had a lower rate but a higher human development index. On the other hand, Orissa and Bihar scored low in both financial inclusion and social development. Manyika et al. (2016) concluded that financial inclusion could achieve financial growth and improve society's overall development. Sharma (2016) discussed the relationship between financial inclusion and economic growth with the Granger causality test, which revealed linkage among outreach and development and deposit and gross domestic product (GDP). Therefore, financial inclusion is connected to economic growth. Kim et al. (2018) studied the link between financial inclusion and socioeconomic development in OIC countries. The results indicated that the financial inclusion and well-being of the nation were connected. Gutiérrez-Nieto and Serrano-Cinca (2019) realized that technology and artificial intelligence are connected and focus on the client's individualization.

Muqtada (2020) found that macroeconomic policymaking would require rethinking to meet the challenges faced due to COVID-19. COVID has resulted in a slowdown in the economy that would require several policy measures to protect lives and livelihoods. It would require an expansionary macroeconomic policy, albeit prudent, that focuses robustly on stepped-up investment, job growth, and reduction of inequality and vulnerability.

4.6. Consumers' Financial Inclusion and Financial Capability

Opening a bank account and savings habits help use the amount during any emergency and are growing among millennials and lower-income groups (Friedline and West 2016; Hogarth and Anguelov 2003). Adams and West (2015) found that savings help utilize money during emergencies among the lower-income group and may find more mechanisms if they adopt financial schemes. Friedline and West (2016) clarifies the stance of financial inclusion and economic inclusion wherein the former focuses entirely upon the usage of banking schemes and the latter is about wealth equity and distribution. Revers (2019) analyzed the importance of the broader financial capability concept in predicting emergency savings for those living above and below the poverty line and suggested that programs aimed at encouraging emergency savings should enhance financial self-efficacy and financial inclusion. Financial inclusion is a fundamental element of the financial capability index (Náñez Alonso et al. 2022; Potocki and Cierpiał-Wolan 2019). (Kempson et al. 2005; Kempson et al. 2017) proposed financial inclusion within three domains: "financial knowledge, financial skills and financial confidence". All the components were ably supported by (Xiao et al. 2015) and were associated with financial behavior with financial capability.

4.7. Socioeconomic Determinants of Financial Inclusion

Socioeconomic Determinants of Financial Inclusion

This subsection directly addresses the current state of understanding regarding the relationship between socioeconomic factors and financial inclusion. It reviews literature that elucidates how variables like income, education, and demographics impact an individual's access to financial services. This section contributes to RQ1 by providing a foundational understanding of the key socioeconomic determinants that influence financial inclusion.

By identifying specific socioeconomic factors that affect financial inclusion, this section also informs RQ2. It highlights areas within socioeconomic development, such as education and income inequality, that require focused attention to enhance financial inclusion.

In discussing existing literature, gaps in the current research become apparent, such as the need for more detailed studies on how specific demographic groups are impacted by financial exclusion. This links to RQ3 by suggesting areas for future research.

Income Levels and Employment Status

Income is a primary determinant of financial inclusion. Higher income levels generally correlate with greater access to financial services. Studies like Demirguc-Kunt et al. (2015) have shown that individuals with steady income streams are more likely to have bank accounts and access credit facilities. Employment status also plays a significant role. Formal employment often provides easier access to financial services through payroll accounts or employer-sponsored financial products.

Education and Financial Literacy

Education is a key factor in enhancing an individual's understanding and usage of financial services. Higher levels of education correlate with increased financial literacy, which in turn leads to better engagement with financial services (Lusardi and Mitchell 2014). Financial literacy campaigns and education programs can significantly boost financial inclusion, especially in regions with low levels of formal education.

Demographic Factors: Age, Gender, and Ethnicity

Age, gender, and ethnicity are important in understanding disparities in financial inclusion. For example, younger and older age groups might face different barriers to access (Campero and Kaiser 2013). Gender disparity in financial inclusion is a significant issue. Women often face higher barriers to access financial services due to factors like legal restrictions or societal norms (Demirguc-Kunt et al. 2013). Ethnic minorities and indigenous populations can also face unique challenges, often stemming from systemic inequalities or discrimination.

Technological Factors and Geographic Location (Urban vs. Rural): Enhancing Financial Inclusion

Geographic location plays a pivotal role in financial inclusion. People in rural areas often have less access to financial services compared to their urban counterparts due to factors like distance from financial institutions and lack of infrastructure (Beck et al. 2007). Technological solutions like mobile banking have begun to bridge this urban-rural divide, as documented in studies of mobile money in African countries.

This subsection contributes to RQ1 by examining how technology has transformed the landscape of financial inclusion, a critical aspect of current understanding. It includes insights on digital banking, mobile money, and fintech innovations. In terms of RQ2, this section points to technological advancements as a crucial area for focused development, emphasizing how technology can bridge socio-economic divides and enhance financial accessibility.

The discussion of technology in financial inclusion also reveals potential research gaps, such as the long-term impact of digital financial services on different socioeconomic groups, aligning with RQ3.

Cultural and Social Norms

Cultural and social norms influence attitudes towards money, savings, and credit, which can affect financial inclusion. In some cultures, traditional forms of saving and lending may be preferred over formal financial services. Trust in financial institutions is another key aspect. In regions where there is a historical mistrust of banking systems, people are less likely to engage with formal financial services.

Government Policies and Regulation

Government policies and regulatory environments can either facilitate or hinder financial inclusion. For instance, policies that encourage branch expansion in underserved areas or mandate financial literacy in school curricula can promote inclusion. Similarly, regulation that lowers entry barriers for non-traditional financial service providers can enhance access and competition.

5. Conclusions

5.1. Financial Inclusion, Fintech, and Artificial Intelligence

The rise of digital initiatives has vastly expanded access to banking services globally. In India, for instance, government initiatives like the Jan Dhan Yojana have significantly driven financial inclusion, offering basic banking services with simplified KYC processes and no minimum balance requirements. Despite these advancements, challenges such as financial illiteracy and technological backwardness persist, hindering the full utilization of financial services by many, especially first-time account holders (Nanda and Kaur 2016). In the BFSI industry, artificial intelligence (AI) and machine learning (ML) have revolutionized efficiency, customer experience, and risk management (Zeinalizadeh et al. 2015; Cao et al. 2022). Developers and relevant companies should highlight the advantages and practical benefits of using Open API (application programming interface) when building applications based on it. This will encourage more businesses and users to use the technology and further progress fintech innovations (Wang 2023).

Fintech's expanding role is crucial in promoting economic growth aligned with Sustainable Development Goal 8 (SDG-8), which focuses on decent work and economic growth. Cryptocurrencies, with appropriate legal and environmental controls, could further facilitate this progress (Abakah et al. 2023). The manuscript underscores the importance of digital banking services and financial literacy in fostering financial inclusion, emphasizing the need for financial education, especially among vulnerable groups.

Artificial Intelligence (AI) has significantly enhanced credit scoring and risk assessment, particularly for individuals with limited credit histories or low incomes. AI-powered identification and verification systems have also improved security while reducing fraud. However, the digital divide remains a significant challenge, as limited access to technology and digital skills prevents some from fully benefiting from AI-driven financial services. Concerns also arise regarding privacy, data security, and the potential for unsustainable debt levels due to easy access to AI-enabled loans (Yanting and Ali 2023). Together, policy-makers, companies, and people need to tackle these issues and make sure that the benefits of automation and artificial intelligence are maximized while limiting any unfavorable effects.

5.2. Financial Literacy of Consumers, Financial Capacity, and Financial Inclusion

Financial inclusion is a key driver of inclusive growth, which is crucial for overall stability. Adequate access to financial products and services for all societal sectors is vital, particularly in emerging economies. Financial literacy generates demand for these services and is a global issue with far-reaching implications.

The current economic downturn has heightened policymakers' focus on financial capability, or the ability of individuals to manage their finances effectively. Studies show significant variability in financial capability across different demographic groups, with young, unemployed single adults often having the lowest financial capability compared to older, employed individuals in dual-income households (Taylor 2011).

5.3. SHGs, Women Empowerment, Gender Issues, and Financial Inclusion

Self-help groups (SHGs) have become globally recognized as microcredit institutions contributing to women's empowerment. Policies promoting savings and loans can strengthen SHGs, particularly in rural areas. Enhancing infrastructure like education and healthcare and overseeing SHG standards are critical for sustainable development. Financial literacy is particularly beneficial for women accessing microfinance (Swamy 2014). Collaboration with banks, corporations, and other agencies is essential to evolve SHGs into member-owned and managed institutions, contributing to financial inclusion and inclusive growth. Financial inclusion programs have shown positive impacts on gender equality and income for impoverished women.

5.4. Intensifying and Inclusivity of Access to Financial Services for Consumers

Being part of the financial system requires access to a variety of financial services and products. These resources address individual financial needs and are crucial for financial inclusion, which is closely linked to poverty alleviation (Sarma and Pais 2011). Despite its importance, financial inclusion has been somewhat overlooked in behavioral science literature. A comprehensive financial system facilitates income generation, leading to opportunities for entrepreneurship and holistic growth.

Financial literacy, digital penetration, AI, women's empowerment, addressing gender issues, and ensuring access to finance are key areas needing attention from policymakers and governments. These factors are vital in achieving the United Nations' Sustainable Development Goals. A robust banking network is instrumental in national development, offering employment, infrastructure, and overall growth. The effectiveness of financial inclusion measures varies across countries, but their potential to significantly improve the socioeconomic conditions in underdeveloped and developing nations is undeniable. Businesses that prioritize moral conduct and community service frequently foster more fulfilling environments for their stakeholders and workers, which raises satisfaction and happiness levels all around. These actions also contribute to the growth of financial inclusion by making it easier for a larger group of people to get financial resources and services (Xi and Wang 2024).

Policy Implications and Frameworks for Financial Inclusion

This section addresses RQ1 by reviewing the role of policy and regulatory frameworks in promoting financial inclusion, an essential aspect of current knowledge. It looks at various government initiatives and international guidelines that shape financial inclusion efforts. For RQ2, it identifies key policy areas that require attention for socioeconomic development, such as regulatory reforms and targeted financial education programs.

Lastly, in relation to RQ3, this subsection can highlight gaps in policy-related research, such as the effectiveness of different regulatory approaches in diverse socio-economic contexts, suggesting directions for future investigation. Financial inclusion is a multifaceted concept with significant implications for socio-economic development. It urges continued research and policy efforts to address the challenges and leverage the opportunities presented by financial inclusion. The research discusses the impact of different regulatory environments on financial inclusion and the importance of data privacy in online financial services. It suggests that policymakers should focus on creating an enabling environment for financial inclusion to thrive. The conclusion also outlines several areas for future research, such as the impact of fintech innovations, the role of financial inclusion in healthcare financing, and its relationship with mental health.

6. Sustainable Financial Inclusion Theory and Future Research Agenda

6.1. Sustainable Financial Inclusion Theory

The concept of financial inclusion is integral to achieving the Sustainable Development Goals (SDGs) and reducing inequality, as emphasized in the United Nations Sustainable Development Agenda 2030. Recognized for its role in promoting economic growth and lessening income disparity, financial inclusion remains a pivotal subject in contemporary research. The theory of sustainable financial inclusion, although lacking a universally accepted definition due to its subjective nature and varying individual and country-specific needs, is universally acknowledged as vital. Basic banking services form the cornerstone of financial inclusion, and initiatives like those implemented by the Indian government for basic accounts are transformative towards achieving inclusivity in India. The proposed theory aims to establish a framework for sustained financial inclusion, highlighting the necessity for accessible, affordable financial products and services for all, ensuring no one is left out. Key to this is building a robust infrastructure network, emphasizing the importance of the availability, affordability, and utilization of financial services.

6.2. Future Research Agenda

The global endeavor towards financial inclusion, while challenging, gains momentum through the endorsement of influential figures and public awareness via financial inclusion quotes. Future research should focus on the following areas:

- Digital Age and Financial Inclusion: Examining the influence of digital banking on unbanked populations and its socio-economic effects.
- Gender and Financial Inclusion: Investigating financial inclusion's role in empowering women and understanding gender-based disparities in financial access.
- **Financial Education and Literacy**: Assessing the socio-economic impacts of financial literacy and understanding financial service adoption through behavioral economics.
- Impact on Policies and Regulations: Evaluating how varying regulatory frameworks affect microfinancing and the socio-economic implications of online financial data privacy.
- **Health and Wellbeing**: Exploring financial inclusion's role in healthcare financing and its relation to mental health.
- Sustainability and Environmental Considerations: Investigating financial inclusion's support for environmentally friendly initiatives and its alignment with the SDGs.
- **Fintech Innovations**: Analyzing the pros and cons of blockchain for financial inclusion and the socio-economic impacts of new fintech products.
- **Sociocultural Variables and Financial Behavior**: Studying the influence of cultural attitudes and trust in institutions on financial service adoption and use.

Theoretical Extension: The sustainable financial inclusion theory proposes a new framework focusing on ensuring financial goods and services are accessible, affordable, and utilized by all societal segments. It advocates for a comprehensive approach with a robust infrastructure and a customer-centric model, particularly addressing the needs of the impoverished in rural areas. The theory encourages empirical research linking financial service access to developmental outcomes and aims to tackle challenges for inclusive growth. Recognizing obstacles to financial inclusion, it seeks solutions for the socioeconomic advancement of diverse social groups.

This research agenda seeks to fill existing knowledge gaps and tackle emerging challenges in finance and socioeconomics, aiming to guide future studies and policy development in financial inclusion.

Author Contributions: Conceptualization, V.K., D.M. and B.S.; methodology, V.K. and N.A.; software, V.K. and B.S.; validation, V.K. and N.A.; formal analysis, D.M. and B.S.; investigation, V.K.; resources, V.K. and B.S.; data curation, V.K. and B.S.; writing—original draft preparation, V.K. and N.A.; writing—review and editing, V.K. and B.S.; visualization, V.K.; supervision, V.K.; project administration, V.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data sharing is not applicable—no new data are generated, or the article describes entirely theoretical research.

Conflicts of Interest: The authors declare that there are no conflicts of interest.

References

Abakah, Emmanuel Joel Aikins, Aviral Kumar Tiwari, Sudeshna Ghosh, and Buhari Doğan. 2023. Dynamic Effect of Bitcoin, Fintech and Artificial Intelligence Stocks on Eco-Friendly Assets, Islamic Stocks and Conventional Financial Markets: Another Look Using Quantile-Based Approaches. *Technological Forecasting and Social Change* 192: 122566. [CrossRef]

Adams, Deborah, and Stacia West. 2015. Asset Building among Low Income Adults: An Exploratory Study with Participants in an Emergency Savings Program. *Journal of Community Practice* 23: 436–61. [CrossRef]

Adetunji, Olubanjo Michael, and Olayinka David-West. 2019. The Relative Impact of Income and Financial Literacy on Financial Inclusion in Nigeria. *Journal of International Development* 31: 312–35. [CrossRef]

Agarwal, Sumit, and Yeow Hwee Chua. 2020. FinTech and Household Finance: A Review of the Empirical Literature. *China Finance Review International* 10: 361–76. [CrossRef]

Ajzen, Icek. 1991. The Theory of Planned Behavior. Organizational Behavior and Human Decision Processes 50: 179–211. [CrossRef]

- Allen, Franklin, Asli Demirguc-Kunt, Leora Klapper, and Maria Martinez Peria. 2016. The Foundations of Financial Inclusion: Understanding Ownership and Use of Formal Accounts. *Journal of Financial Intermediation* 27: 1–30. [CrossRef]
- Alliance for Financial Inclusion. 2019. Alliance for Financial Inclusion Policy Model: AFi Core Set of Financial Inclusion Indicators. *Bringing Smart Policies to Life* 1–26. Available online: www.afi-global.org (accessed on 14 March 2022).
- Almansour, Mohammed. 2023. Artificial Intelligence and Resource Optimization: A Study of Fintech Start-Ups. *Resources Policy* 80: 103250. [CrossRef]
- Alnemer, Hashem Abdullah. 2022. Determinants of Digital Banking Adoption in the Kingdom of Saudi Arabia: A Technology Acceptance Model Approach. *Digital Business* 2: 100037. [CrossRef]
- An, Chaofan, Xiaoxia He, and Lin Zhang. 2023. The Coordinated Impacts of Agricultural Insurance and Digital Financial Inclusion on Agricultural Output: Evidence from China. *Heliyon* 9: e13546. [CrossRef] [PubMed]
- Andreou, Panayiotis C., and Sofia Anyfantaki. 2021. Financial Literacy and Its Influence on Internet Banking Behavior. *European Management Journal* 39: 658–74. [CrossRef]
- Ardic, Oya Pinar, Kathryn Imboden, and Alexia Latortue. 2013. Financial Access 2012 Getting to a More Comprehensive Picture. Available online: https://www.cgap.org/sites/default/files/cgap_forum_FAS2012.pdf (accessed on 14 March 2022).
- Arner, Douglas W., Dirk A. Zetzsche, Ross P. Buckley, and Janos N. Barberis. 2019. The Identity Challenge in Finance: From Analogue Identity to Digitized Identification to Digital KYC Utilities. *European Business Organization Law Review* 20: 55–80. [CrossRef]
- Arner, Douglas W., Ross P. Buckley, Dirk A. Zetzsche, and Robin Veidt. 2020. Sustainability, FinTech and Financial Inclusion. *European Business Organization Law Review* 21: 7–35. [CrossRef]
- Asongu, Simplice A., Joseph Nnanna, and Paul N. Acha-Anyi. 2020. Inequality and Gender Economic Inclusion: The Moderating Role of Financial Access in Sub-Saharan Africa. *Economic Analysis and Policy* 65: 173–85. [CrossRef]
- Babajide, Abiola A., Folasade B. Adegboye, and Alexander E. Omankhanlen. 2015. Financial Inclusion and Economic Growth in Nigeria. *International Journal of Economics and Financial Issues* 5: 629–37.
- Baidoo, Samuel Tawiah, Elliot Boateng, and Mary Amponsah. 2018. Understanding the Determinants of Saving in Ghana: Does Financial Literacy Matter? *Journal of International Development* 30: 886–903. [CrossRef]
- Bajunaied, Kholoud, Nazimah Hussin, and Suzilawat Kamarudin. 2023. Behavioral Intention to Adopt FinTech Services: An Extension of Unified Theory of Acceptance and Use of Technology. *Journal of Open Innovation: Technology, Market, and Complexity* 9: 100010. [CrossRef]
- Banerjee, Sayantani, and Greeshma Francis. 2014. Financial Inclusion and Social Development. *International Journal of Scientific Research and Management* 2321: 13–18.
- Bansal, Harvir S., and Shirley F. Taylor. 2002. Investigating Interactive Effects in the Theory of Planned Behavior in a Service-Provider Switching Context. *Psychology & Marketing* 19: 407–25. [CrossRef]
- Barik, Rajesh, and Pritee Sharma. 2019. Analyzing the Progress and Prospects of Financial Inclusion in India. *Journal of Public Affairs* 19: e1948. [CrossRef]
- Beck, Thorsten, Asli Demirguc-Kunt, and Maria Soledad Martinez Peria. 2007. Reaching out: Access to and Use of Banking Services across Countries. *Journal of Financial Economics* 85: 234–66. [CrossRef]
- Bhanot, Disha, Varadraj Bapat, and Bera Sasadhar. 2012. Studying Financial Inclusion in North-East India. *International Journal of Bank Marketing* 30: 465–84. [CrossRef]
- Campero, Alejandra, and Karen Kaiser. 2013. *Access to Credit: Awareness and Use of Formal and Informal Credit Institutions*. Working Papers, No. 2013-07. Ciudad de México: Banco deMéxico. Available online: https://econpapers.repec.org/RePEc:bdm:wpaper:2013-07 (accessed on 10 February 2021).
- Cao, Ting, Wade D. Cook, and M. Murat Kristal. 2022. Has the Technological Investment Been Worth It? Assessing the Aggregate Efficiency of Non-Homogeneous Bank Holding Companies in the Digital Age. *Technological Forecasting and Social Change* 178: 121576. [CrossRef]
- Chukhrova, Nataliya, and Arne Johannssen. 2021. Stochastic Claims Reserving Methods with State Space Representations: A Review. *Risks* 9: 198. [CrossRef]
- Claessens, Stijn, and Enrico Perotti. 2007. Finance and Inequality: Channels and Evidence. *Journal of Comparative Economics* 35: 748–73. Clámara, Noelia, Ximena Peña, and Tuesta David. 2014. Factors That Matter for Financial Inclusion: Evidence from Peru. In *BBVA Research*, Working Paper No. 14/09. Madrid: BBVA Research, pp. 1–26. [CrossRef]
- Cude, Brenda J., Swarn Chatterjee, and Jamal Tavosi. 2019. Financial Knowledge among Iranian Investors. *International Journal of Consumer Studies* 43: 503–13. [CrossRef]
- Demirguc-Kunt, Asli, Leora Klapper, and Dorothe Singer. 2013. Financial Inclusion and Legal Discrimination against Women Evidence from Developing Countries. Policy Research Working Paper 6416. Available online: https://www.mfw4a.org/sites/default/files/resources/Financial%20Inclusion%20and%20Legal%20Discrimination%20Against%20Women%20Evidence%20from%20 Developing%20Countries.pdf (accessed on 10 February 2021).
- Demirguc-Kunt, Asli, Leora Klapper, Dorothe Singer, and Peter Van Oudheusden. 2015. The Global Findex Database 2014 Measuring Financial Inclusion around the World. Policy Research Working Paper 7255. Available online: https://documents1.worldbank.org/curated/en/187761468179367706/pdf/WPS7255.pdf (accessed on 10 February 2021).

- Demirguc-Kunt, Asli, Leora Klapper, Dorothe Singer, Saniya Ansar, and Jake Hess. 2018. *The Global Findex Database* 2017: *Measuring Financial Inclusion and the Fintech Revolution*. Washington, DC: International Bank for Reconstruction and Development/The World Bank. [CrossRef]
- Denyer, David, and David Tranfield. 2009. Producing a Systematic Review. In *The Sage Handbook of Organizational Research Methods*. Thousand Oaks: Sage Publications Ltd.
- Didenko, Iryna, Jolanta Paucz-Olszewska, Serhiy Lyeonov, Anna Ostrowska-Dankiewicz, and Zbigniew Ciekanowski. 2020. Social Safety and Behavioral Aspects of Populations Financial Inclusion: A Multicountry Analysis. *Journal of International Studies* 13: 347–59. [CrossRef]
- Diniz, Eduardo, Rene Birochi, and Marlei Pozzebon. 2012. Triggers and Barriers to Financial Inclusion: The Use of ICT-Based Branchless Banking in an Amazon County. *Electronic Commerce Research and Applications* 11: 484–94. [CrossRef]
- Duvendack, Maren, and Philip Mader. 2020. Impact of Financial Inclusion in Low- and Middle-Income Countries: A Systematic Review of Reviews. *Journal of Economic Surveys* 34: 594–629. [CrossRef]
- East, Robert. 1993. Investment Decisions and the Theory of Planned Behaviour. *Journal of Economic Psychology* 14: 337–75. [CrossRef] Erlando, Angga, Feri Dwi Riyanto, and Someya Masakazu. 2020. Financial Inclusion, Economic Growth, and Poverty Alleviation: Evidence from Eastern Indonesia. *Heliyon* 6: e05235. [CrossRef]
- Ferrari, Ms Valeria. 2022. The Platformisation of Digital Payments: The Fabrication of Consumer Interest in the EU FinTech Agenda. *Computer Law and Security Review* 45: 105687. [CrossRef]
- Friedline, Terri, and Stacia West. 2016. Financial Education Is Not Enough: Millennials May Need Financial Capability to Demonstrate Healthier Financial Behaviors. *Journal of Family and Economic Issues* 37: 649–71. [CrossRef]
- Garikipati, Supriya. 2008. The Impact of Lending to Women on Household Vulnerability and Women's Empowerment: Evidence from India. *World Development* 36: 2620–42. [CrossRef]
- George, Okello Candiya Bongomin, and Ntayi Joseph Mpeera. 2020. Mobile Money Adoption and Usage and Financial Inclusion: Mediating Effect of Digital Consumer Protection. *Digital Policy, Regulation and Governance* 22: 157–76. [CrossRef]
- Goodell, John, Satish Kumar, Weng Marc Lim, and Debidutta Pattnaik. 2021. Artificial Intelligence and Machine Learning in Finance: Identifying Foundations, Themes, and Research Clusters from Bibliometric Analysis. *Journal of Behavioral and Experimental Finance* 32: 100577. [CrossRef]
- Goyal, Kirti, and Satish Kumar. 2020. Financial Literacy: A Systematic Review and Bibliometric Analysis. *International Journal of Consumer Studies* 45: 80–105. [CrossRef]
- Goyal, Sandeep, Amit Kapoor, and Bruno Sergi. 2020. Empowering Rural Women through Shared Value Approach Study of GNFC Neem Project in India. *World Review of Entrepreneurship Management and Sustainable Development* 16: 359–79. [CrossRef]
- Grohmann, Antonia, Theres Klühs, and Lukas Menkhoff. 2018. Does Financial Literacy Improve Financial Inclusion? Cross Country Evidence. *World Development* 111: 84–96. [CrossRef]
- Grohs-Müller, Stefan, and Bettina Greimel-Fuhrmann. 2018. Students' Money Attitudes and Financial Behaviour: A Study on the Relationship between Two Components of Financial Literacy. *Empirische Pädagogik* 32: 369–86.
- Gutiérrez-Nieto, Begoña, and Carlos Serrano-Cinca. 2019. 20 Years of Research in Microfinance: An Information Management Approach. *International Journal of Information Management* 47: 183–97. [CrossRef]
- Han, Rui, and Maring Melecky. 2013. Financial Inclusion for Stability: Access to Bank Deposits and the Deposit Growth during the Global Financial Crisis. Policy Research Working Paper 6577. Available online: https://mpra.ub.uni-muenchen.de/48339/(accessed on 10 February 2021).
- Hanafizadeh, Payam, and Ahad Zare Ravasan. 2018. A Model for Selecting IT Outsourcing Strategy: The Case of e-Banking Channels. *Journal of Global Information Technology Management* 21: 111–38. [CrossRef]
- Hassan, Md Sharif, Md Aminul Islam, Mohd Faizal Yusof, and Hussen Nasir. 2023. Users' Fintech Services Acceptance: A Cross-Sectional Study on Malaysian Insurance & Takaful Industry. *Heliyon* 9: e21130. [CrossRef]
- Hogarth, Jeanne M., and Chris E. Anguelov. 2003. Can the Poor Save? Journal of Financial Counseling and Planning 14: 1–18.
- Hsu, Tsuen-ho, Yi-sheng Wang, and Su-chan Wen. 2006. Using the Decomposed Theory of Planning Behavioural to Analyse Consumer Behavioural Intention towards Mobile Text Message Coupons. *Journal of Targeting, Measurement and Analysis for Marketing* 14: 309–24. [CrossRef]
- Hussain, Shahzad, Raazia Gul, and Sabeeh Ullah. 2023. Role of Financial Inclusion and ICT for Sustainable Economic Development in Developing Countries. *Technological Forecasting and Social Change* 194: 122725. [CrossRef]
- Iqbal, Badar Alam, and Shaista Sami. 2017. Papel de Los Bancos En La Inclusión Financiera En La India. *Contaduria y Administracion* 62: 644–56. [CrossRef]
- Jappelli, Tullio, and Mario Padula. 2013. Investment in Financial Literacy and Saving Decisions. *Journal of Banking & Finance* 37: 2779–92. [CrossRef]
- Karlan, Dean, and Jonathan Zinman. 2010. Expanding Credit Access: Using Randomized Supply Decisions to Estimate the Impacts. *Review of Financial Studies* 23: 433–64. [CrossRef]
- Kass-Hanna, Josephine, Angela C. Lyons, and Fan Liu. 2022. Building Financial Resilience through Financial and Digital Literacy in South Asia and Sub-Saharan Africa. *Emerging Markets Review* 51: 100846. [CrossRef]
- Kawimbe, Sidney. 2020. An Assessment of the Impact of Mobile Financial Services on Financial Inclusion and Economic Development in Zambia. *International Journal of Current Research* 12: 10122–31. [CrossRef]

- Kempson, Elaine, Sharon Collard, and Nick Moore. 2005. Measuring Financial Capability: An Exploratory Study. Consumer Research 37, Financial Services Authority. Available online: http://www.fsa.gov.uk/pubs/consumer-research/crpr37.pdf (accessed on 18 March 2021).
- Kempson, Elaine, Andrea Finney, and Christian Poppe. 2017. Financial Well-Being A Conceptual Model and Preliminary Analysis. Available online: https://www.researchgate.net/publication/318852257_Financial_Well%20Being_A_Conceptual_Model_and_Preliminary_Analysis (accessed on 18 March 2021).
- Kim, Dai Won, Jung Suk Yu, and M. Kabir Hassan. 2018. Financial Inclusion and Economic Growth in OIC Countries. *Research in International Business and Finance* 43: 1–14. [CrossRef]
- Kitchenham, Barbara, O. Pearl Brereton, David Budgen, Mark Turner, John Bailey, and Stephen Linkman. 2009. Systematic Literature Reviews in Software Engineering—A Systematic Literature Review. *Information and Software Technology* 51: 7–15. [CrossRef]
- Klapper, Leora, Mayada El-Zoghbi, and Jake Hess. 2016. Achieving the Sustainable Development Goals The Role of Financial Inclusion. *CGAP* 23: 2016.
- Kochar, Anjini, Closepet Nagabhushana, Ritwik Sarkar, Rohan Shah, and Geeta Singh. 2022. Financial Access and Women's Role in Household Decisions: Empirical Evidence from India's National Rural Livelihoods Project. *Journal of Development Economics* 155: 102821. [CrossRef]
- Koh, Francis, Kok Fai Phoon, and Cao Duy Ha. 2018. Digital Financial Inclusion in South East Asia. In *Handbook of Blockchain, Digital Finance, and Inclusion*, 1st ed. Amsterdam: Elsevier Inc., vol. 2. [CrossRef]
- Koomson, Isaac, and Muazu Ibrahim. 2018. Financial Inclusion and Growth of Non-Farm Enterprises in Ghana. In *Financing Sustainable Development in Africa*. Edited by Uchenna R. Efobi and Simplice Asongu. Cham: Springer International Publishing, pp. 369–96. [CrossRef]
- Koomson, Isaac, Renato A. Villano, and David Hadley. 2020. Intensifying Financial Inclusion through the Provision of Financial Literacy Training: A Gendered Perspective. *Applied Economics* 52: 375–87. [CrossRef]
- Kraus, Sascha, Matthias Breier, and Sonia Dasí-Rodríguez. 2020. The Art of Crafting a Systematic Literature Review in Entrepreneurship Research. *International Entrepreneurship and Management Journal* 16: 1023–42. [CrossRef]
- Kumar, Nitin. 2013. Financial Inclusion and Its Determinants: Evidence from India. *Journal of Financial Economic Policy* 5: 4–19. [CrossRef]
- Lauer, Kate, and Timothy Lyman. 2015. *Digital Financial Inclusion: Implications for Customers, Regulators, Supervisors, and Standard-Setting Bodies*. Cgap Brief. Washington, DC: The World Bank.
- Levine, Ross. 2005. Finance and Growth: Theory and Evidence. Handbook of Economic Growth 1: 865–934. [CrossRef]
- Lusardi, Annamaria, and Olivia Mitchell. 2014. The Economic Importance of Financial Literacy: Theory and Evidence. *Journal of Economic Literature* 52: 5–44. [CrossRef]
- Lusardi, Annamaria, and Peter Tufano. 2015. Debt Literacy, Financial Experiences, and Over Indebtedness. *Journal of Pension Economics and Finance* 14: 332–68. [CrossRef]
- Lusardi, Annamaria, Olivia S. Mitchell, and Vilsa Curto. 2014. Financial Literacy and Financial Sophistication in the Older Population. *Journal of Pension Economics and Finance* 13: 347–66. [CrossRef]
- Mahmud, Khaled, Md. Mahbubul Alam Joarder, and Kazi Sakib. 2023. Customer Fintech Readiness (CFR): Assessing Customer Readiness for Fintech in Bangladesh. *Journal of Open Innovation: Technology, Market, and Complexity* 9: 100032. [CrossRef]
- Maity, Shrabanti. 2023. Self Help Groups, Microfinance, Financial Inclusion and Social Exclusion: Insight from Assam. *Heliyon* 9: e16477. [CrossRef]
- Malik, Shruti, Girish Chandra Maheshwari, and Archana Singh. 2020. Formal vs. Informal Demand for Credit by Women in Urban India: A Comparative Study in Slums of Delhi and Mumbai. *Gender in Management* 36: 131–48. [CrossRef]
- Manyika, James, Susan Lund, Marc Singer, Olivia White, and Chris Berry. 2016. *Digital Finance for All: Powering Inclusive Growth in Emerging Economies*. New York: McKinsey Global Institute. Available online: https://www.mckinsey.com/~/media/McKinsey/Featured%20Insights/Employment%20and%20Growth/How%20digital%20finance%20could%20boost%20growth%20in%20 emerging%20economies/MGI-Digital-Finance-For-All-Executive-summary-September-2016.ashx (accessed on 20 February 2022).
- Marangunić, Nikola, and Andrina Granić. 2015. Technology Acceptance Model: A Literature Review from 1986 to 2013. *Universal Access in the Information Society* 14: 81–95. [CrossRef]
- Mi Alnaser, Feras, Samar Rahi, Mahmoud Alghizzawi, and Abdul Hafaz Ngah. 2023. Does Artificial Intelligence (AI) Boost Digital Banking User Satisfaction? Integration of Expectation Confirmation Model and Antecedents of Artificial Intelligence Enabled Digital Banking. *Heliyon* 9: e18930. [CrossRef]
- Morgan, Peter J., and Trinh Quang Long. 2020. Financial Literacy, Financial Inclusion, and Savings Behavior in Laos. *Journal of Asian Economics* 68: 101197. [CrossRef]
- Morgan, Peter, and Long Trinh. 2019. Determinants and Impacts of Financial Literacy in Cambodia and Viet Nam. *Journal of Risk and Financial Management* 12: 19. [CrossRef]
- Mouna, Amari, and Anis Jarboui. 2016. Financial Literacy in Tunisia: Its Determinants and Its Implications on Investment Behavior. *Research in International Business and Finance* 39: 568–77. [CrossRef]
- Moutinho, Luiz, and Anne Smith. 2000. Modelling Bank Customer Satisfaction through Mediation of Attitudes towards Human and Automated Banking. *International Journal of Bank Marketing* 18: 124–34. [CrossRef]

- Muqtada, Muhammed. 2020. *Macroeconomic Policy, Price Stability and Inclusive Growth BT—Quest for Inclusive Growth in Bangladesh: An Employment-Focused Strategy*. Edited by Muhammed Muqtada. Singapore: Springer, pp. 93–135. [CrossRef]
- Nanda, Kajole, and Mandeep Kaur. 2016. Financial Inclusion and Human Development: A Cross-Country Evidence. *Management and Labour Studies* 41: 127–53. [CrossRef]
- Náñez Alonso, Sergio L., Javier Jorge-Vazquez, Miguel Ángel Echarte Fernández, Konrad Kolegowicz, and Wojciech Szymla. 2022. Financial Exclusion in Rural and Urban Contexts in Poland: A Threat to Achieving SDG Eight? *Land* 11: 539. [CrossRef]
- Ozili, Peterson K. 2020. Contesting Digital Finance for the Poor. Digital Policy, Regulation and Governance 22: 135–51. [CrossRef]
- Paramasivan, Chelliah, and V. Ganeshkumar. 2013. Overview of Financial Inclusion in India. *International Journal of Management and Development Studies* 2: 45–49.
- Park, Cyn-Young, and Rogelio Mercado. 2018. Financial inclusion, poverty, and income inequality. *The Singapore Economic Review* 63: 185–206. [CrossRef]
- Pasali, Selahattin Selsah. 2013. Where Is the Cheese? Synthesizing a Giant Literature on Causes and Consequences of Financial Sector Development. In *Policy Research Working Paper No. 6655, The World Bank Finance and Private Sector Development*. Geneva: Graduate Institute of International Studies.
- Pattnaik, Debidutta, Sougata Ray, and Raghu Raman. 2024. Applications of Artificial Intelligence and Machine Learning in the Financial Services Industry: A Bibliometric Review. *Heliyon* 10: e23492. [CrossRef]
- Paul, Justin, and Alexander Rosado-Serrano. 2019. Gradual Internationalization vs. Born-Global/International New Venture Models: A Review and Research Agenda. *International Marketing Review* 36: 830–58. [CrossRef]
- Paul, Justin, Arun Mittal, and Garima Srivastav. 2016. Impact of Service Quality on Customer Satisfaction in Private and Public Sector Banks. *International Journal of Bank Marketing* 34: 606–22. [CrossRef]
- Paul, Justin, Sundar Parthasarathy, and Parul Gupta. 2017. Exporting Challenges of SMEs: A Review and Future Research Agenda. *Journal of World Business* 27: 327–42. [CrossRef]
- Pnevmatikakis, Aristodemos, Stathis Kanavos, George Matikas, Konstantina Kostopoulou, Alfredo Cesario, and Sofoklis Kyriazakos. 2021. Risk Assessment for Personalized Health Insurance Based on Real-World Data. *Risks* 9: 46. [CrossRef]
- Polychroniou, Panagiotis V., and Ioannis Giannikos. 2009. A Fuzzy Multicriteria Decision-making Methodology for Selection of Human Resources in a Greek Private Bank. *Career Development International* 14: 372–87. [CrossRef]
- Potocki, Tomasz, and Marek Cierpiał-Wolan. 2019. Factors Shaping the Financial Capability of Low-Income Consumers from Rural Regions of Poland. *International Journal of Consumer Studies* 43: 187–98. [CrossRef]
- Rastogi, Shailesh, and E. Ragabiruntha. 2018. Financial Inclusion and Socioeconomic Development: Gaps and Solution. *International Journal of Social Economics* 45: 1122–40. [CrossRef]
- Reserve Bank of India. 2015. Report of the Committee on Medium-Term Path on Financial Inclusion. Kolkata: Reserve Bank of India.
- Reyers, Michelle. 2019. Financial Capability and Emergency Savings among South Africans Living above and below the Poverty Line. *International Journal of Consumer Studies* 43: 335–47. [CrossRef]
- Sapre, Nikhil. 2023. Financial Inclusion: Philosophical and Methodological Underpinnings. *Qualitative Research in Financial Markets* 15: 445–52. [CrossRef]
- Sarma, Mandira, and Jesim Pais. 2011. Financial Inclusion and Development. *Journal of International Development* 23: 613–28. [CrossRef] Sharma, Dipasha. 2016. Nexus between Financial Inclusion and Economic Growth: Evidence from the Emerging Indian Economy. *Journal of Financial Economic Policy* 8: 13–36. [CrossRef]
- Sharma, Dipasha, Sonali Bhattacharya, and Shagun Thukral. 2018. Assessment of Financial Inclusive Policy in Indian Economy. *International Journal of Ethics and Systems* 34: 304–20. [CrossRef]
- Söderberg, Inga Lill. 2016. Gender Stereotyping in Financial Advisors' Assessment of Customers. *Journal of Financial Services Marketing* 17: 259–72. [CrossRef]
- Swamy, Vighneswara. 2014. Financial Inclusion, Gender Dimension, and Economic Impact on Poor Households. *World Development* 56: 1–15. [CrossRef]
- Tappi, Marco, Gianluca Nardone, and Fabio Gaetano Santeramo. 2022. On the Relationships among Durum Wheat Yields and Weather Conditions: Evidence from Apulia Region, Southern Italy. *Bio-Based and Applied Economics* 11: 123–30. [CrossRef]
- Taylor, Mark. 2011. Measuring Financial Capability and Its Determinants Using Survey Data. *Social Indicators Research* 102: 297–314. [CrossRef]
- Tinker, I. 2001. Poverty and Gender in Developing Nations. In *International Encyclopedia of the Social & Behavioral Sciences*. Heights: Waveland Press, pp. 11904–7. [CrossRef]
- Tranfield, David, David Denyer, and Palminder Smart. 2003. Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review. *British Journal of Management* 14: 207–22. [CrossRef]
- Vorobyev, Ivan, and Anna Krivitskaya. 2022. Reducing False Positives in Bank Anti-Fraud Systems Based on Rule Induction in Distributed Tree-Based Models. *Computers & Security* 120: 102786. [CrossRef]
- Wang, Jen Sheng. 2022. Verification Techniques in FinTech Compared from User Perspectives. *Social Science Computer Review* 41: 1438–55. [CrossRef]
- Wang, Jen Sheng. 2023. Reconfigure and Evaluate Consumer Satisfaction for Open API in Advancing FinTech. *Journal of King Saud University—Computer and Information Sciences* 35: 101738. [CrossRef]

Wann, Christi R. 2017. The Impact of Teaching Financial Literacy to College Students. *Journal of Economics and Finance Education* 16: 98–109.

World Bank. 2018. Financial Inclusion Financial Inclusion Is a Key Enabler to Reducing Poverty and Boosting Prosperity. Available online: https://www.worldbank.org/en/topic/financialinclusion/overview (accessed on 18 December 2022).

Xi, Wenzhi, and Yingdong Wang. 2023. Digital Financial Inclusion and Quality of Economic Growth. Heliyon 9: e19731. [CrossRef]

Xi, Haomeng, and Jizhou Wang. 2024. Social Governance, Family Happiness, and Financial Inclusion. *Finance Research Letters* 61: 104946. [CrossRef]

Xiao, Jing Jian. 2008. Applying Behavior Theories to Financial Behavior. In *Handbook of Consumer Finance Research*. Edited by Jing Jian Xiao. New York: Springer, pp. 69–81. [CrossRef]

Xiao, Jing Jian, Cheng Chen, and Lei Sun. 2015. Age Differences in Consumer Financial Capability. *International Journal of Consumer Studies* 39: 387–95. [CrossRef]

Yadav, Priyanka, and Anil Kumar Sharma. 2016. Financial Inclusion in India: An Application of TOPSIS. *Humanomics* 32: 328–51. [CrossRef]

Yanting, Zou, and Muhammad Ali. 2023. Artificial Intelligence, Digital Finance, and Financial Inclusion: A Conceptual Framework. In *Financial Inclusion across Asia: Bringing Opportunities for Businesses*. Edited by Choi-Meng Leong, Muhammad Ali, Syed Ali Raza, Chin-Hong Puah and Ibrahim Halil Eksi. Bradford: Emerald Publishing Limited, pp. 77–85. [CrossRef]

Zeinalizadeh, Nooshin, Amir Abbas Shojaie, and Mohammad Shariatmadari. 2015. Modeling and Analysis of Bank Customer Satisfaction Using Neural Networks Approach. *International Journal of Bank Marketing* 33: 717–32. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.